---
title: "P-Series: Programme"
description: "International agreements and programme documentation"
---

import { LinkCard, CardGrid, Badge } from '@astrojs/starlight/components';

International agreements and programme documentation

## Documents (15 total)

| Document | Description | Version | Date |
|----------|-------------|---------|------|
| [DoI-China-EN](/cospas-sarsat/p-series/doi-china-en/) | Declaration Of Intent |  | July 1988 |
| [P001](/cospas-sarsat/p-series/p001/) | The International Cospas-Sarsat Programme Agreement |  | October 1984 |
| [P002](/cospas-sarsat/p-series/p002/) | Procedure For The Notification Of Association With The International Cospas-Sars |  | December 1992 |
| [P005](/cospas-sarsat/p-series/p005/) | P5NOV05.doc |  | April 2005 |
| [P006](/cospas-sarsat/p-series/p006/) | P6NOV05E.doc |  | May 2005 |
| [P007](/cospas-sarsat/p-series/p007/) | EN | Issue 5 | February 2019 |
| [P008](/cospas-sarsat/p-series/p008/) | Arrangement On Cooperation Between The Cooperating Agencies Of The Parties To Th |  | July 1988 |
| [P009](/cospas-sarsat/p-series/p009/) | Understanding Between The States Parties To The International Cospas-Sarsat Prog |  | July 1988 |
| [P010-Rev3](/cospas-sarsat/p-series/p010-rev3/) | _LIST OF STATES & ORGANIZATIONS | Issue 1 | February 2026 |
| [P011](/cospas-sarsat/p-series/p011/) | C/S P.011 Issue 2 Rev.7 | Issue 2 | October 2025 |
| [P012](/cospas-sarsat/p-series/p012/) | P12NOV05.doc | Issue 1 | November 2005 |
| [P014](/cospas-sarsat/p-series/p014/) | P014-Final with signatures |  | July 1988 |
| [P015](/cospas-sarsat/p-series/p015/) | C/S P.015 - Issue 2 Rev.1 | Issue 2 | October 2024 |
| [P016](/cospas-sarsat/p-series/p016/) | Cospas-Sarsat | Issue 1 | December 2016 |
| [P017](/cospas-sarsat/p-series/p017/) | Declaration Of Intent |  | July 1988 |

